Author's response to reviews

Title: Pilot case-control investigation of risk factors for hip fractures in the urban Indian population

Authors:

Ruchira M Jha (ruchirajha@gmail.com)
Ambrish Mithal (ambrishmital@rediffmail.com)
Nidhi Malhotra (docnidhi1@rediffmail.com)
Edward M Brown (embrown@rics.bwh.harvard.edu)

Version: 6 Date: 24 February 2010

Author's response to reviews:

Major revisions required -

1. Tables: tables must be created using a Tables tool, and should not be linked files, figures, or generated with tabbed text. Tables must be divided into rows and columns and be oriented in a portrait page layout. Please ensure that text alignment in columns is clear and consistent. We recommend the MS Word auto-format template: simple 1. Please do not include vertical lines in your tables.

Tables have been recreated using Tables tool and MS Word auto format- Simple 1 template used.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

DONE

Thanks